Author's response to reviews

Title: Immunohistochemical localization of urokinase-type plasminogen activator, urokinase-type plasminogen activator receptor and alpha2-antiplasmin in human corneal perforation: a case report

Authors:

Koji Sugioka (sugioka@med.kindai.ac.jp)
Aya Kodama (akodama@med.kindai.ac.jp)
Koji Yoshida (kojiy@med.kindai.ac.jp)
Kiyotaka Okada (kiyokada@med.kindai.ac.jp)
Masahiko Fukuda (fukuda-m@med.kindai.ac.jp)
Yoshikazu Shimomura (yoshis@med.kindai.ac.jp)

Version: 3 Date: 19 November 2012

Author's response to reviews:

November 19, 2012
Ms. April Gerobin
The BioMed Central Editorial Team

RE: MS: 2357038507481614

Dear Ms. Gerobin,

Thank you for your e-mail of November 13, 2012 regarding our manuscript entitled “Immunohistochemical localization of urokinase-type plasminogen activator, urokinase type-plasminogen activator receptor and alpha2-antiplasmin in human corneal perforation: a case report”.

We are pleased to learn that you are prepared to accept our manuscript after the English has been revised. We thank the referees for their time and effort in reviewing our manuscript and providing valuable comments.

According to the indication by Referee 2 and upon the recommendation of the BMC Editorial Team, we submitted our article to Edanz for English editing and proofreading. A native-English speaker with scientific expertise edited our article.

We have corrected the revised manuscript and the corrections are underlined in our revised manuscript.

Thank you for your efforts on our behalf.

Sincerely yours,

Koji Sugioka MD
Department of Ophthalmology
Kinki University Faculty of Medicine
377-2 Ohno-Higashi, Osaka-Sayama City, Osaka, 589-8511 Japan
E-mail: sugioka@ med.kindai.ac.jp